Author's response to reviews

Title: Identification of prognostic factors and surgical indications for metastatic gastric cancer.

Authors:

Yasuhiko Mohri (ya-mohri@clin.medic.mie-u.ac.jp)
Koji Tanaka (gouji@clin.medic.mie-u.ac.jp)
Masaki Ohi (mohi1012@clin.medic.mie-u.ac.jp)
Susumu Saigusa (saigusas@yahoo.co.jp)
Hiromi Yasuda (yasuda19792@yahoo.co.jp)
Yuji Toiyama (ytoi0725@clin.medic.mie-u.ac.jp)
Toshimitsu Araki (taraki@clin.medic.mie-u.ac.jp)
Yasuhiro Inoue (yasinoue@clin.medic.mie-u.ac.jp)
Masato Kusunoki (kusunoki@clin.medic.mie-u.ac.jp)

Version: 2
Date: 27 January 2014

Author's response to reviews:

We made the change our manuscript.
We added the section of the Acknowledgements as below.

Acknowledgements
We would like to thank Masato Okigami, Tadanobu Shimura, Yuki Imaoka, Satoru Kondo, and Takahito Kitajima for participation in data collection. This study was performed without funding from grants or (industrial) sponsors.